Author’s response to reviews

Title: The expression of microRNA-375 in plasma and tissue is matched in human colorectal cancer

Authors:

Lingling Xu (xulingling05@sina.com)
Minzhe Li (leeminzhe@hotmail.com)
Min Wang (damy1988@163.com)
Dong Yan (yd15yt88@163.com)
Guosheng Feng (fgs010bjcyh@126.com)
Guangyu An (anguangyu@hotmail.com)

Version: 5 Date: 3 September 2014

Author’s response to reviews: see over
Dear editor,

This is a re-submission of MS: 1750449064129322. Thank you very much for your effort to review the revised version of the manuscript and the acceptance for BMC Cancer journal publication. Taking into consideration of the final reviewers’ comments carefully and in order to obtain a better understanding of the expression levels of miRs in CRC tissues and matched normal mucosa samples, Figure 2 was changed by using $2^{-\Delta\Delta CT}$ instead of $2^{-\Delta CT}$. The relative explanations were added in the section of materials and methods on page 7, line 201-202. We are going to study the molecular mechanism of miR-375 in the subsequent research.

We hope that the revised manuscript conforms to the publication in BMC Cancer journal and look forward to hearing from you soon.

Thank you and best regards.

Yours sincerely,

Guangyu An  M.D., Ph.D.

Department of Oncology,

Beijing Chao-Yang Hospital, Capital Medical University, Beijing 100020, China,

E-mail addresses: anguangyu@hotmail.com,

Tel:+86 1085231569

Fax:+86 1085231407